RESEARCH PAPER



Topology optimization framework for structures subjected to stationary stochastic dynamic loads

Fernando Gomez¹ D · Billie F. Spencer Jr¹

Received: 3 March 2018 / Revised: 14 September 2018 / Accepted: 18 September 2018 / Published online: 29 September 2018 © Springer-Verlag GmbH Germany, part of Springer Nature 2018

Abstract

The field of topology optimization has progressed substantially in recent years, with applications varying in terms of the type of structures, boundary conditions, loadings, and materials. Nevertheless, topology optimization of stochastically excited structures has received relatively little attention. Most current approaches replace the dynamic loads with either equivalent static or harmonic loads. In this study, a direct approach to problem is pursued, where the excitation is modeled as a stationary zero-mean filtered white noise. The excitation model is combined with the structural model to form an augmented representation, and the stationary covariances of the structural responses of interest are obtained by solving a Lyapunov equation. An objective function of the optimization scheme is then defined in terms of these stationary covariances. A fast large-scale solver of the Lyapunov equation is implemented for sparse matrices, and an efficient adjoint method is proposed to obtain the sensitivities of the objective function. The proposed topology optimization framework is illustrated for four examples: (i) minimization of the displacement of a mass at the free end of a cantilever beam subjected to a stochastic dynamic tip load, (iii) minimization of tip displacement and acceleration of a cantilever beam subjected to a stochastic dynamic tip load, (iii) minimization of a plate subjected to multiple stochastic dynamic loads. The results presented herein demonstrate the efficacy of the proposed approach for efficient multi-objective topology optimization of stochastically excited structures, as well as multiple input-multiple output systems.

Keywords Topology optimization · Stochastic dynamics · Lyapunov equation · Finite element · Multi-objective optimization

1 Introduction

Traditional structural design methods are based on an iterative procedure that focuses on structural safety; however, such designs are generally not optimal (Xu et al. 2017a, b). In this regard, structural optimization is an important design tool that can consider various performance objectives. Within this field, topology optimization seeks

Responsible Editor: Byeng D Youn

 Fernando Gomez gmzsnch2@illinois.edu
 Billie F. Spencer Jr bfs@illinois.edu to obtain optimal material layout according to a objective function subjected to given design constraints (Bendsøe and Sigmund 2003). Extensive research has been done in topology optimization to develop well-posed formulations (Bendsøe and Kikuchi 1988; Kohn and Strang 1986; Bendsøe and Sigmund 1999; Sigmund and Petersson 1998; Sigmund 2007) and solve inherent numerical problems such as mesh dependency, checkerboard patterning, islanding, and local minima (Díaz and Sigmund 1995; Sigmund and Petersson 1998).

Topology optimization has been successfully applied to solve the minimum compliance problem subjected to deterministic static loading for general structures (Bendsøe and Sigmund 2003; Talischi et al. 2012), as well as domains representing buildings (Beghini et al. 2014). It also has been applied extensively to dynamic problems, such as eigenfrequency optimization for free vibration (Olhoff 1976, 1989; Filipov et al. 2016), minimum dynamic compliance for harmonic dynamic vibration (Ma et al. 1995;

¹ Department of Civil and Environmental Enginnering, University of Illinois at Urbana-Champaign, Urbana, IL 61801, USA

Filipov et al. 2016), and strain energy optimization for transient vibration (Kang et al. 2006; Behrou and Guest 2017). The latter research usually employs time-domain solutions, which is extremely time consuming, to obtain the response of the structure and the sensitivities; some researchers have recently proposed schemes to reduce the computational time of this step using mode acceleration model reduction methods (Zhao and Wang 2016), or multiresolution approaches (Filipov et al. 2016).

Many of the most severe loads that structures withstand are stochastic in nature: earthquake, wind, snow, rain, ocean waves, jet noise, turbulence in the boundary layer, etc. (Soong and Grigoriu 1993; Li and Chen 2009). Nevertheless, structural optimization of stochastically excited structures has developed more slowly than its deterministic counterpart; for example, some work has been done in preliminary size optimization of aircraft for coupled aerodynamic structural response under stochastic gust loading (Fidkowski et al. 2008). Most of the research has been done in size optimization of buildings using Monte Carlo simulations (Balling et al. 2009); however, numerous simulations are required to obtain meaningful results, which makes this approach time prohibitive. Researchers in earthquake engineering have also considered specific ground motion records (Allahdadian and Boroomand 2016); in this case, each record is only one realization of the underlying random process. Xu et al. (2017a, b) proposed a parametric size optimization method for linear and nonlinear buildings subjected to stationary and nonstationary stochastic excitation by solving a small-scale Lyapunov equation problem.

Chun et al. (2016) proposed a reliability-based topology optimization framework for a stationary Gaussian random process excitation using a discrete representation of the excitation and solving in the time-domain; this approach provides reasonable results, but the time stepping algorithm requires considerable computational effort. Recently, researchers have proposed formulations to minimize the covariance of the displacement of a fixed point (Zhang et al. 2015; Hu et al. 2016; Zhu et al. 2017; Yang et al. 2017) or multiple points (Spencer et al. 2016; Gomez and Spencer 2017). Others (e.g., Zhang et al. 2015, Zhu et al. 2017, Yang et al. 2017) used frequency-domain solutions based on the first few frequencies and mode shapes of the structure to obtain the covariance of the displacement. The computational time increases considerably when more modes are used, and the numerical errors in the sensitivities of the mode shapes increase (Haftka and Adelman 1989). Hu et al. (2016) used an explicit time-domain solution to obtain the covariance of displacements, which provides reasonable results, but involves two time-domain analyses. In addition, all previous approaches focus on displacement response and cannot easily be generalized to multi-objective problems due to requiring additional modes to achieve adequate accuracy, or multiple input-multiple output systems. Spencer et al. (2016) and Gomez and Spencer (2017) used the Lyapunov equation to obtain the covariance of the response; however, numerical difficulties in solving the Lyapunov equation limited the size of the meshes that could be employed.

The solution of the Lyapunov equation (i.e., $AX + XA^{T} +$ $\mathbf{Q} = \mathbf{0}$) is typically obtained using the Bartels-Steward algorithm or the Hessenberg-Schur algorithm (Golub et al. 1979), both of which require the Schur factorization of the matrix A. Variations of these algorithms are implemented in typical software for scientific computing such as Matlab and Python. These methods provide good results for small dense matrices A, Q; however, two practical issues arise for large-scale systems: (i) they require $\mathbb{O}(N^3)$ floatingpoint operations and (ii) $\mathbb{O}(N^2)$ memory (Kressner 2008), both of which impose a significant constraint on the size of the problem using current computers. Several approaches have been developed that exploit the low-rank nature of the matrix **Q** using Krylov subspace methods (Saad 1990; Jbilou and Riquet 2006; Kressner 2008) or the matrix sign function decomposition with Newton's iterative method (Higham 2008; Balzer 1980). These algorithms reduce the computational time and required memory; however, they fail when the symmetric part of A, i.e., $A + A^{T}$ is not negative definite (Benner et al. 2008), which is the standard case for state space matrices for structural systems. The Alternating Direction Implicit (ADI) iteration algorithm was developed to solve linear systems in terms of optimal shift parameters (Wachspress 2013) and has been adapted to solve the Lyapunov and algebraic Riccati equations with low-rank matrices **Q** with superlinear convergence (Penzl 1999; Li and White 2002; Benner et al. 2008). This method is adopted herein to efficiently solve the Lyapunov equation with large-scale sparse matrices.

This paper proposes a new multi-objective topology optimization framework for stochastically excited structures. The performance function is given in terms of the stationary response covariances, obtained by solving a large-scale Lyapunov equation. The proposed performance function is versatile and allows optimization of the structure's displacement, velocity, and/or acceleration, including multiobjective functions; also, an efficient adjoint method is developed to obtain the sensitivities of this performance function, which accommodates the use of gradient-based updating procedures. This paper is organized as follows: Section 2 describes the problem formulation including the state space representation of the structure and excitation, the response under stochastic excitation, and topology optimization formulation; Section 3 provides details for the solution of the optimization problem, including large-scale solution of Lyapunov equations, sensitivity analysis, and enforcing symmetry constraints; Section 4 shows numerical examples of the optimization of a cantilever beam subjected to a stochastic dynamic tip load and a cantilever beam subjected to a stochastic dynamic base excitation; Section 5 presents the conclusions of this work.

2 Problem formulation

This section formulates the topology optimization problem for structures subjected to stationary stochastic dynamic loading. First, the structural system is converted from the standard second-order differential equation into the state space representation, and the excitation is modeled as a filtered white noise. Subsequently, the covariance matrix of the stationary stochastic responses is obtained. Finally, the topology optimization framework is presented based on this formulation.

2.1 State space representation

Consider a dynamic linear system with N degrees of freedom (DOF) whose equation of motion is given by

$$\mathbf{M}\ddot{\mathbf{u}} + \mathbf{C}\dot{\mathbf{u}} + \mathbf{K}\mathbf{u} = \mathbf{G}\mathbf{f}(t) \tag{1}$$

where **M**, **C**, and **K** represent the mass, damping, and stiffness matrices, respectively; **G** is the load effect matrix; $\mathbf{f}(t)$ is the input excitation vector; and **u** is the displacement vector.

Defining the vector \mathbf{x}_s as

$$\mathbf{x}_{s} = \begin{bmatrix} \mathbf{u}^{\mathrm{T}} & \dot{\mathbf{u}}^{\mathrm{T}} \end{bmatrix}^{\mathrm{T}}$$
(2)

the system can be represented in the state space form by

$$\dot{\mathbf{x}}_{s} = \mathbf{A}_{s}\mathbf{x}_{s} + \mathbf{B}_{s}\mathbf{f}(t)$$

$$\mathbf{y} = \mathbf{C}_{s}\mathbf{x}_{s} + \mathbf{D}_{s}\mathbf{f}(t)$$
 (3)

where the state matrices A_s and B_s are

$$\mathbf{A}_{s} = \begin{bmatrix} \mathbf{0}_{N \times N} & \mathbf{I}_{N \times N} \\ -\mathbf{M}^{-1}\mathbf{K} & -\mathbf{M}^{-1}\mathbf{C} \end{bmatrix}, \quad \mathbf{B}_{s} = \begin{bmatrix} \mathbf{0}_{N \times 1} \\ \mathbf{M}^{-1}\mathbf{G} \end{bmatrix}$$
(4)

with **0** is a matrix of zeros, and **I** is the identity matrix, with dimensions given by the subscripts; **y** is the vector of output responses of interest corresponding to the matrices C_s and D_s .

2.2 Stochastic excitation

Many physical phenomena can be modeled in terms of a stochastic process (Soong and Grigoriu 1993). In this study, the excitation is assumed to be a zero-mean stationary random process modeled as a filtered white-noise with the following space state space representation

$$\dot{\mathbf{x}}_{f} = \mathbf{A}_{f}\mathbf{x}_{f} + \mathbf{B}_{f}\mathbf{w}(t)$$

$$\mathbf{f} = \mathbf{C}_{f}\mathbf{x}_{f}$$
 (5)

where the matrices \mathbf{A}_{f} , \mathbf{B}_{f} , and \mathbf{C}_{f} are based on the characteristics of the excitation; \mathbf{x}_{f} is the state vector of the excitation model with N_{f} states; and $\mathbf{w}(t)$ is a vectored white noise process.

The white noise vector $\mathbf{w}(t)$ satisfies

$$\mathbb{E}(\mathbf{w}(t)) = 0, \quad \mathbb{E}(\mathbf{w}(t_1)\mathbf{w}(t_2)) = 2\pi \mathbf{S}_0 \delta(t_1 - t_2) \tag{6}$$

where $\mathbb{E}(\cdot)$ is the expected value operator, \mathbf{S}_0 is the magnitude two-sided constant $m \times m$ power spectral density matrix, and $\delta(\cdot)$ is the Dirac delta function.

An augmented state vector \mathbf{x}_a can be defined as

$$\mathbf{x}_{a} = \begin{bmatrix} \mathbf{x}_{s}^{\mathrm{T}} & \mathbf{x}_{f}^{\mathrm{T}} \end{bmatrix}^{\mathrm{T}}$$
(7)

yielding an augmented system whose state space representation is given by

$$\dot{\mathbf{x}}_{a} = \mathbf{A}_{a}\mathbf{x}_{a} + \mathbf{B}_{a}\mathbf{w}(t) \mathbf{y} = \mathbf{C}_{a}\mathbf{x}_{a}$$
(8)

where the matrices A_a , B_a , and C_a are given by

$$\mathbf{A}_{a} = \begin{bmatrix} \mathbf{A}_{s} & \mathbf{B}_{s} \mathbf{C}_{f} \\ \mathbf{0}_{N_{f} \times N} & \mathbf{A}_{f} \end{bmatrix}, \quad \mathbf{B}_{a} = \begin{bmatrix} \mathbf{0}_{2N \times 1} \\ \mathbf{B}_{f} \end{bmatrix}, \qquad (9)$$
$$\mathbf{C}_{a} = \begin{bmatrix} \mathbf{C}_{s} & \mathbf{D}_{s} \mathbf{C}_{f} \end{bmatrix}$$

Note that the input of the augmented system is a vectored white noise, and the output of the augmented system is the outputs of interest for the structural system.

2.3 Structural response

The covariance matrix of the response of a linear time invariant system subjected to a white noise excitation, such as the one considered in this study, can be computed directly from Soong and Grigoriu (1993)

$$\dot{\mathbf{\Gamma}}_{\mathbf{x}_{a}} = \mathbf{A}_{a} \mathbf{\Gamma}_{\mathbf{x}_{a}} + \mathbf{\Gamma}_{\mathbf{x}_{a}} \mathbf{A}_{a}^{\mathrm{T}} + 2\pi \mathbf{B}_{a} \mathbf{S}_{0} \mathbf{B}_{a}^{\mathrm{T}}$$
(10)

where

$$\boldsymbol{\Gamma}_{\mathbf{x}_{a}} = \mathbb{E}((\mathbf{x}_{a} - \boldsymbol{\mu}_{\mathbf{x}_{a}})(\mathbf{x}_{a} - \boldsymbol{\mu}_{\mathbf{x}_{a}})^{\mathrm{T}}) = \mathbb{E}(\mathbf{x}_{a}\mathbf{x}_{a}^{\mathrm{T}})$$
(11)

Assuming that the input white noise has a zero mean and that $\Gamma_{\mathbf{x}_a}(0) = \mathbf{0}$, then the mean value of the response is also zero, i.e., $\mu_{\mathbf{x}_a} = \mathbf{0}$, then $\Gamma_{\mathbf{x}_a} = \mathbb{E}(\mathbf{x}_a \mathbf{x}_a^T)$. By considering the stationary part of the response, the covariance matrix becomes constant and its time derivative becomes zero, which yields the Lyapunov equation

$$\mathbf{A}_{\mathbf{a}}\boldsymbol{\Gamma}_{\mathbf{x}_{\mathbf{a}}} + \boldsymbol{\Gamma}_{\mathbf{x}_{\mathbf{a}}}\mathbf{A}_{\mathbf{a}}^{\mathrm{T}} + 2\pi \,\mathbf{B}_{\mathbf{a}}\mathbf{S}_{\mathbf{0}}\mathbf{B}_{\mathbf{a}}^{\mathrm{T}} = \mathbf{0}$$
(12)

The Lyapunov equation possesses a unique solution if the matrix A_a is Hurwitz, which means that its eigenvalues have strictly negative real parts. In the case considered herein, the eigenvalues of A_a are equal to the eigenvalues of A_s and A_g , due to the block structure of the former matrix (Strang 2003). The eigenvalues of A_s and A_g will have negative real parts if the structure and the excitation are strictly stable.

Also, the matrix $2\pi \mathbf{B}_{a} \mathbf{S}_{0} \mathbf{B}_{a}^{T}$ is symmetric positive semidefinite, if \mathbf{B}_{a} has linearly independent columns and $2\pi \mathbf{S}_{0}$ is positive definite. Finally, because of the properties of the covariance matrix, the solution of the Lyapunov equation is guaranteed to be symmetric positive semi-definite.

The covariance of the structural output \mathbf{y} can be calculated via

$$\boldsymbol{\Gamma}_{\boldsymbol{y}} = \mathbb{E}(\boldsymbol{y}\boldsymbol{y}^{T}) = \boldsymbol{C}_{a}\mathbb{E}(\boldsymbol{x}_{a}\boldsymbol{x}_{a}^{T})\boldsymbol{C}_{a}^{T} = \boldsymbol{C}_{a}\boldsymbol{\Gamma}_{\boldsymbol{x}_{a}}\boldsymbol{C}_{a}^{T}$$
(13)

2.4 Topology optimization formulation

Design variables in continuous-domain topology optimization are chosen as the relative density in each element (Bendsøe and Sigmund 2003). Therefore for element *n*, the relative density variable is denoted by z_n , where $n \in$ {1, 2, ..., N_{el} }, and N_{el} is the total number of elements. The optimization formulation is thus given by:

Find $\mathbf{z} = [z_1, z_2, \dots, z_N]$ such that:

$$\min_{\mathbf{z}} J(\mathbf{z}) = \phi \left(\Gamma_{\mathbf{x}_{a}}(\mathbf{z}), \mathbf{z} \right)$$

s.t. $g(\mathbf{z}) = V(\mathbf{z}) - V_{\max} \le 0$
 $\mathbf{A}_{a} \Gamma_{\mathbf{x}_{a}} + \Gamma_{\mathbf{x}_{a}} \mathbf{A}_{a}^{\mathrm{T}} + 2\pi \mathbf{B}_{a} \mathbf{S}_{0} \mathbf{B}_{a}^{\mathrm{T}} = \mathbf{0}$
 $z_{n} \in [z_{\min}, z_{\max}] \text{ for } n = 1, 2, \dots, N_{\mathrm{el}}$ (14)

where ϕ is a continuous differentiable positive scalar function, $\Gamma_{\mathbf{x}_a}$ is the stationary covariance of the response, V is the volume, V_{max} is the volume constraint, and z_{min} and z_{max} are the lower and upper bounds on the density variables.

Note that the proposed performance function allows consideration of many different problems. To illustrate this point for different types of response, consider the following particular case

$$J(\mathbf{z}) = \mathbf{F}(\mathbf{z}) : \mathbf{\Gamma}_{\mathbf{X}_a}(\mathbf{z})$$
(15)

where $J(\mathbf{z})$ is the objective function, : represents the double dot product between matrices or the sum of the diagonal entries of their product, **F** is a symmetric positive semidefinite matrix. Some examples are shown next, for which the specific matrix **F** is described (see the Appendix for the details of the derivations):

1) Covariance of the various DOFs defined by the output equation $\mathbf{y} = \mathbf{C}_{a}\mathbf{x}_{a}$,

$$J = \mathbf{C}_{\mathbf{a}} \mathbf{\Gamma}_{\mathbf{x}_{\mathbf{a}}} \mathbf{C}_{\mathbf{a}}^{\mathrm{T}}, \quad \mathbf{F} = \mathbf{C}_{\mathbf{a}}^{\mathrm{T}} \mathbf{C}_{\mathbf{a}}$$
(16)

For example, $\mathbf{C}_{a} = \mathbf{C}_{u} = \begin{bmatrix} \mathbf{I}_{N \times N} & \mathbf{0}_{N \times N} & \mathbf{0}_{N \times N_{f}} \end{bmatrix}$ for displacements, $\mathbf{C}_{a} = \mathbf{C}_{\mathbf{B}} = \begin{bmatrix} \mathbf{0}_{N \times N} & \mathbf{I}_{N \times N} & \mathbf{0}_{N \times N_{f}} \end{bmatrix}$ for velocities, and $\mathbf{C}_{\mathbf{ii}} = \begin{bmatrix} -\mathbf{M}^{-1}\mathbf{K} & -\mathbf{M}^{-1}\mathbf{C} & \mathbf{M}^{-1}\mathbf{G}\mathbf{C}_{f} \end{bmatrix}$ for accelerations. In each of these cases, all, or a selected number, of the DOFs can be employed in the objective function. 2) Expected value of the static compliance, based on the static case of potential energy minimization

$$J = \mathbb{E}(\mathbf{u}^{\mathrm{T}}\mathbf{K}\mathbf{u}), \quad \mathbf{F} = \mathbf{C}_{\mathbf{u}}^{\mathrm{T}}\mathbf{K}\mathbf{C}_{\mathbf{u}}$$
(17)

3) Expected value of the kinetic energy

$$J = \mathbb{E}(\dot{\mathbf{u}}^{\mathrm{T}}\mathbf{M}\mathbf{R}), \quad \mathbf{F} = \mathbf{C}_{\dot{\mathbf{u}}}^{\mathrm{T}}\mathbf{K}\mathbf{C}_{\mathbf{R}}$$
(18)

 Linear combination of performance functions with non-negative coefficient α. For example, engineering designers may desire to perform multi-objective optimization to minimize displacements and accelerations (Xu et al. 2017a, b). In this case,

$$J = \mathbf{F}_1 : \mathbf{\Gamma}_{\mathbf{x}_a} + \alpha \mathbf{F}_2 : \mathbf{\Gamma}_{\mathbf{x}_a}, \quad \mathbf{F} = \mathbf{F}_1 + \alpha \mathbf{F}_2$$
(19)
where $\mathbf{F}_1 = \mathbf{C}_{\mathbf{u}}^{\mathrm{T}} \mathbf{C}_{\mathbf{u}}, \mathbf{F}_2 = \mathbf{C}_{\mathbf{u}}^{\mathrm{T}} \mathbf{C}_{\mathbf{u}}.$

As illustrated here, the performance function is completely defined by the covariance of the response Γ_{x_a} , which is obtained through solution of the Lyapunov equation. Consequently, the stochastic optimization problem has been transformed into a deterministic counterpart.

A gradient-based optimization procedure is preferred, such as the method of moving asymptotes (Svanberg 1987), for which the gradient of the performance function and constraints are required. Section 3.3 provides details on how to obtain them.

3 Solution method

This section describes the numerical details for the solution of the topology optimization problem formulation for stochastic excitations. These details include the solution of the Lyapunov equation, the efficient evaluation of sensitivities, symmetry constraints to reduce problem size, and optimization details.

3.1 Material interpolation

The well-known continuous approach, based on intermediate element densities, is adopted to obtain the optimal topology (Bendsøe and Sigmund 2003). For each element n, a relative density variable z_n is chosen, where $n \in$ $\{1, 2, ..., N_{el}\}$. Then, Young's modulus and density for each element are obtained by some interpolation rule; in this work, SIMP interpolation is used (Bendsøe and Sigmund 1999). However, in dynamic problems, local spurious modes may appear due to the artificially low ratios of stiffness to mass using SIMP; to overcome this problem, a modified SIMP is implemented where the elements with low relative density are penalized with a larger exponent to compute the mass (Tcherniak 2002; Du and Olhoff 2007). Du and Olhoff (2007) recommend using the penalization factor equal to p + 3 for elements with density below 0.1, and they propose corrections to ensure continuity and differentiability of the interpolation rule. Section 4.1 presents an example and further discussion of the influence of this local modes penalization on optimal topologies.

In this study, the following relationships are used

$$E(z) = [\epsilon + (1 - \epsilon)z^{p}]E^{0}$$

$$\rho(z) = \begin{cases} z^{q}\rho^{0}, & \text{if } z \ge 0.1 \\ c_{0}z^{p+3}\rho^{0}, & \text{if } z < 0.1 \end{cases}$$
(20)

where E and ρ are Young's modulus and density for the element with variable z, E^0 and ρ^0 are Young's modulus and density for the solid material, p > 1 and $q \ge 1$ are the penalization factors, ϵ is a small number (Talischi et al. 2012), and $c_0 = 10^{p+3-q}$ is a coefficient to ensure continuity in density interpolation rule. Alternative interpolation rules to address this issue can be considered (Zhu et al. 2010, 2018).

3.2 Lyapunov equation solver

As seen in Section 2.3, the covariance of the response of the system can be determined from the solution of the Lyapunov equation (see (12)), which is rewritten next for simplicity

$$\mathbf{A}_{a}\boldsymbol{\Gamma}_{\mathbf{x}_{a}} + \boldsymbol{\Gamma}_{\mathbf{x}_{a}}\mathbf{A}_{a}^{\mathrm{T}} + \tilde{\mathbf{B}}_{a}\tilde{\mathbf{B}}_{a}^{\mathrm{T}} = \mathbf{0}$$
(21)

where $\mathbf{A}_a \in \mathbb{R}^{N' \times N'}$ and $\mathbf{\tilde{B}}_a = \sqrt{2\pi} \mathbf{\tilde{S}} \in \mathbb{R}^{N' \times m}$ with $m \ll N'$, and $\mathbf{\tilde{S}}$ is the lower Cholesky factor of \mathbf{S}_0 . The third term of the LHS in this equation is a low-rank matrix, and the state matrix has an indefinite symmetric part; therefore, the Cholesky factor alternating direction implicit (CF-ADI) iterative algorithm is the only suitable method for the solution (Li and White 2002; Benner et al. 2008).

The CF-ADI algorithm solves for the complex matrix $\mathbf{Z} \in \mathbb{C}^{N' \times mn}$

$$\mathbf{Z} = \begin{bmatrix} \mathbf{V}_1 & \mathbf{V}_2 & \dots & \mathbf{V}_n \end{bmatrix}$$
(22)

where the unknown matrix $\Gamma_{\mathbf{x}_a}$ is given by

$$\Gamma_{\mathbf{x}_a} = \mathbf{Z}\mathbf{Z}^{\mathbf{H}} \tag{23}$$

and ^H denotes the Hermitian transpose operation.

The matrices V_i are obtained using the iterative procedure showed in the following equations

$$\mathbf{V}_{1} = \sqrt{-2\Re(p_{1})} (\mathbf{A}_{a} + p_{1}\mathbf{I})^{-1} \tilde{\mathbf{B}}_{a}$$
$$\mathbf{V}_{k} = \sqrt{\frac{\Re(p_{k})}{\Re(p_{k-1})}} [\mathbf{V}_{k-1} - (p_{k} + p_{k-1}^{*})(\mathbf{A}_{a} + p_{k}\mathbf{I})^{-1}\mathbf{V}_{k-1}]$$
(24)

where $\mathbb{R}(\cdot)$ denotes the real part function, * denotes the complex conjugate, and $p_i \in \mathbb{C}^-$ are complex shift parameters with negative real part that satisfy

$$\{p_1, p_2, \dots, p_l\} = \operatorname*{argmin}_{\{p_1, \dots, p_l\} \in \mathbb{C}^-} \max_{t \in \sigma(\mathbf{A}_a)} \frac{|r_l(t)|}{|r_l(-t)|}$$
(25)

817

with $r_l(t) = \prod_{i=1}^{l} (t - p_i)$ and $\sigma(\mathbf{A}_a)$ denotes the spectrum of \mathbf{A}_a . Obtaining these optimal shift parameters is a expensive task (Benner et al. 2008); therefore, in this study, an algorithm to obtain a set of sub-optimal parameters is used instead (Penzl 1999).

To assess the accuracy of the proposed solver with the traditional Bartel-Stewards algorithm, which is a built-in function in Matlab, the objective function for the same problem is computed using both approaches. A uniform rectangular domain of 6×12 discretized into 3200 elements is considered; this smaller problem is chosen due to the excessive computational time required for the traditional solver in larger problems. The absolute difference between the proposed approach and the standard Matlab solver for the objective function is 3.11×10^{-9} and the relative difference is 1.72×10^{-8} ; therefore, the proposed solver achieves an accuracy similar to the traditional solver, and in this example, it requires only 1/643 of the computational time.

The CF-ADI algorithm requires four parameters: k^+ is the number of Ritz values using power iteration, k^- is the number of Ritz values using inverse iteration, l is the number of parameters, and n is the number of iterations; the accuracy and efficiency of the method depend on these parameters. To obtain appropriate values for these parameters, a numerical test was conducted for a rectangular domain of 5×15 discretized into 7500 elements. To assess the accuracy of the algorithm, the residual of the equation is defined as

$$\mathbf{R} = \mathbf{A}_{\mathbf{a}} \mathbf{\Gamma}_{\mathbf{x}_{\mathbf{a}}} + \mathbf{\Gamma}_{\mathbf{x}_{\mathbf{a}}} \mathbf{A}_{\mathbf{a}}^{\mathrm{T}} + \tilde{\mathbf{B}}_{\mathbf{a}} \tilde{\mathbf{B}}_{\mathbf{a}}^{\mathrm{T}}$$
(26)

which should be a matrix of small numbers. Figure 1a shows the logarithm of the Frobenius norm of the residuals, varying the number of parameters and iterations. Figure 1b shows the logarithm of the Frobenius norm of the residual varying the number of Ritz values using power iteration and Ritz values using inverse iteration. As these figures show, the CF-ADI algorithm achieves good results for a sufficiently large number of iterations and parameters; moreover, using more Ritz values with inverse iteration improves accuracy. For a problem of this size with the largest number of iterations and parameters, a direct implementation of this solver requires a computational time of 4.9 s in a computer with processor Intel Xeon E3-1285 v6 @4.10 GHz and 32 Gb of RAM. For comparison using the same computer and the same structure, a timedomain solution of a harmonic excitation of 30 Hz with sampling frequency of 100 Hz and 1024 time steps using an efficient implementation of the Newmark method requires 3.27 s. These timing results show that a complete stochastic solution can be obtained for about the same cost as 1.5 time-history solutions of one harmonic excitation, which demonstrate the efficacy of the proposed approach.



Fig. 1 log $||R||_F$ varying **a** the number of parameters *l* and the number of iterations *n* and **b** the number Ritz values using power iteration k^+ and the number Ritz values using inverse iteration k^-

3.3 Sensitivity analysis

A gradient-based optimization procedure is preferred for computational efficiency purposes; therefore, the gradient of the performance function and constraints are required. The constraints are linear, making the gradient straightforward to obtain. Because the performance function defined in (14a) depends on the covariance matrix, which is implicitly defined by (21), a direct differentiation approach would be expensive due to the large number of variables. Therefore, an adjoint method is proposed, for which the following Lagrangian function is defined with a symmetric positive semi-definite Lagrange multiplier matrix Λ

$$\mathcal{L}(\mathbf{z}, \mathbf{\Lambda}) = J(\mathbf{z}) + \mathbf{\Lambda} : \left(\mathbf{A}_{a}\mathbf{\Gamma}_{\mathbf{x}_{a}} + \mathbf{\Gamma}_{\mathbf{x}_{a}}\mathbf{A}_{a}^{\mathrm{T}} + \tilde{\mathbf{B}}_{a}\tilde{\mathbf{B}}_{a}^{\mathrm{T}}\right)$$
(27)

The corresponding adjoint equation can be solved to obtain Λ , thus eliminating the implicitly defined gradients of Γ_{x_a} ,

$$\mathbf{A}_{a}^{\mathrm{T}}\mathbf{\Lambda} + \mathbf{\Lambda}\mathbf{A}_{a} + \frac{\partial\phi}{\partial\mathbf{\Gamma}_{\mathbf{x}_{a}}} = \mathbf{0}$$
(28)

The previous result is consistent with a similar approach implemented in the optimization of a control cost function (Yan et al. 2016). Equation (28) is a Lyapunov equation, which has a unique solution because A_a^T is Hurwitz. Finally, the sensitivity of the performance function, which is equal to the sensitivity of the Lagrangian function, is given by the following equation

$$\frac{\partial J}{\partial z_n} = \left(\frac{\partial \mathbf{A}_a}{\partial z_n} \mathbf{\Gamma}_{\mathbf{X}_a} + \mathbf{\Gamma}_{\mathbf{X}_a} \frac{\partial \mathbf{A}_a^{\mathrm{T}}}{\partial z_n} + \frac{\partial (\tilde{\mathbf{B}}_a \tilde{\mathbf{B}}_a^{\mathrm{T}})}{\partial z_n} \right) : \mathbf{\Lambda} + \frac{\partial \phi}{\partial z_n}$$
(29)

For the particular case shown in (15), the previous equations reduce to

$$\mathbf{A}_{\mathbf{a}}^{\mathrm{T}}\mathbf{\Lambda} + \mathbf{\Lambda}\mathbf{A}_{\mathbf{a}} + \mathbf{F} = \mathbf{0}$$
(30)

$$\frac{\partial J}{\partial z_n} = \left(\frac{\partial \mathbf{A}_a}{\partial z_n} \mathbf{\Gamma}_{\mathbf{x}_a} + \mathbf{\Gamma}_{\mathbf{x}_a} \frac{\partial \mathbf{A}_a^{\mathrm{T}}}{\partial z_n} + \frac{\partial (\tilde{\mathbf{B}}_a \tilde{\mathbf{B}}_a^{\mathrm{T}})}{\partial z_n} \right) : \mathbf{\Lambda} + \frac{\partial \mathbf{F}}{\partial z_n} : \mathbf{\Gamma}_{\mathbf{x}_a}$$
(31)

Because **F** is symmetric positive semi-definite, the solution of the Lyapunov equation is symmetric positive semi-definite **A**, as assumed previously. Also, note that **F** can be expressed as a low-rank matrix product; hence, the efficient algorithm described in Section 3.2 can be applied.

The sensitivities of the performance function require the derivatives of the matrices A_a , B_a , and F that were defined explicitly in previous sections.

Adjoint sensitivities were successfully confirmed by comparison to finite difference. To assess the efficiency of the proposed method, several runs with different number of elements were performed. Figure 2 shows the computational time for different approaches, demonstrating that the proposed approach (i.e., using the adjoint method with the CF-ADI solver) requires considerably less time.

The detailed derivations of the results presented in this section are provided in the Appendix.

3.4 Symmetry constraints

In many problems, the domain, boundary conditions, and dynamic loadings are spatially symmetric, for example, the building subjected to a base motion in Fig. 3, and consequently, the structural response is constrained to follow symmetric and anti-symmetric modes. In the



Fig. 2 Computational time (FD, finite differences; DD, direct differentiation; AM, adjoint method with native solver; A-ADI, adjoint method with CF-ADI solver)

example shown in Fig. 3, the symmetry in the domain, the boundary conditions, and the loading dictates that the lateral displacement is symmetric, the vertical response is antisymmetric, and the rotation angle response is symmetric. Therefore, using symmetry to reduce the computational burden can be done without loss of generality. Note that the proposed method does not require use of the symmetry constraint.

The symmetry in the response can be enforced using a constraint of the type

$$\mathbf{u} = \mathbf{T}\tilde{\mathbf{u}} \tag{32}$$

where **u** is the vector of all DOFs, **T** is a transformation matrix, and $\tilde{\mathbf{u}}$ is the vector of master DOFs. Moreover, the transformation matrix and vector of all DOFs can be divided



Fig. 3 Example of a symmetric domain with symmetric random loading $% \left({{{\mathbf{F}}_{\mathbf{r}}}_{\mathbf{r}}} \right)$

in blocks, possibly permuted due to DOF numbering, as follows

$$\mathbf{T} = \begin{bmatrix} \mathbf{I} \\ \mathbf{S} \end{bmatrix}, \quad \mathbf{u} = \begin{bmatrix} \tilde{\mathbf{u}} \\ \mathbf{u}^{\mathbf{c}} \end{bmatrix}$$
(33)

where **I** is the identity matrix, **S** is the constraint matrix, and $\mathbf{u}^{\mathbf{c}}$ is the vector of constrained DOFs. The constraint matrix **S** is a sparse matrix with exactly one non-zero entry per row and whose non-zero entries are equal to 1 or -1, depending respectively on whether the constraint is associated with a symmetric or antisymmetric slave DOF. Therefore, the transformation matrix can be written as blocks of identity matrices, negative identity matrices, and zero matrices of consistent dimensions, possibly permuted due to DOF numbering. Similar relations are used for velocities and accelerations.

The reduced system matrices can be obtained through the following well-known relations

$$\tilde{\mathbf{M}} = \mathbf{T}^{\mathrm{T}}\mathbf{M}\mathbf{T}, \ \tilde{\mathbf{C}} = \mathbf{T}^{\mathrm{T}}\mathbf{C}\mathbf{T}
\tilde{\mathbf{K}} = \mathbf{T}^{\mathrm{T}}\mathbf{K}\mathbf{T}, \ \tilde{\mathbf{G}} = \mathbf{T}^{\mathrm{T}}\mathbf{G}$$
(34)

The state space representation of this system and the augmented can be obtained using the reduced system matrices, and the Lyapunov equation can be solved to obtain the covariance of the response $\tilde{\Gamma}_{x_a}$. From the numerical point of view, the implementation can be done efficiently by using the previous relations, because all the matrices are highly sparse. Moreover, this transformation reduces the size of the problem to a approximately half, which results in a corresponding reduction in computation time for a given mesh size.

The covariance matrix Γ_{x_a} of the initial augmented system can be recovered from the covariance matrix $\tilde{\Gamma}_{x_a}$ of the reduced system as follows

$$\boldsymbol{\Gamma}_{\boldsymbol{x}_{a}} = \mathbb{E}(\boldsymbol{x}_{a}\boldsymbol{x}_{a}^{T}) = \boldsymbol{T}\mathbb{E}(\tilde{\boldsymbol{x}}_{a}\tilde{\boldsymbol{x}}_{a}^{T})\boldsymbol{T}^{T} = \boldsymbol{T}\tilde{\boldsymbol{\Gamma}}_{\boldsymbol{x}_{a}}\boldsymbol{T}^{T}$$
(35)

Therefore, using the general form of the transformation matrix yields

$$\Gamma_{\mathbf{x}_{a}} = \begin{bmatrix} \mathbf{I} \\ \mathbf{S} \end{bmatrix} \tilde{\Gamma}_{\mathbf{x}_{a}} \begin{bmatrix} \mathbf{I} & \mathbf{S}^{\mathsf{T}} \end{bmatrix} \implies \Gamma_{\mathbf{x}_{a}} = \begin{bmatrix} \tilde{\Gamma}_{\mathbf{x}_{a}} & \tilde{\Gamma}_{\mathbf{x}_{a}} \mathbf{S}^{\mathsf{T}} \\ \mathbf{S}\tilde{\Gamma}_{\mathbf{x}_{a}} & \mathbf{S}\tilde{\Gamma}_{\mathbf{x}_{a}} \mathbf{S}^{\mathsf{T}} \end{bmatrix}$$
(36)

For the case when all the constrained DOFs have a symmetric response, i.e., $\mathbf{S} = \mathbf{I}$, the cross-covariance equals the auto-covariance, and the correlation coefficient is equal to 1 for all master-constrained pairs. On the contrary, for the case when all the constrained DOFs have an anti-symmetric response, i.e., $\mathbf{S} = -\mathbf{I}$, the cross-covariance equals the negative of the auto-covariance, and the correlation coefficient is equal to -1 for all master-constrained pairs. Typically, some combination of symmetric and antisymmetric constraints is required.

The solution of the proposed optimization problem is summarized in Fig. 4. In the initialization step, the domain is meshed, the element matrices using solid material are computed, the matrices for the excitation model are constructed, and the initial values for the design variables are chosen. Additionally, to avoid mesh dependency and numerical instabilities such as checkerboard patterns and islanding, a filter is applied to the sensitivities (Sigmund and Petersson 1998). A linear hat filter is implemented through a filter matrix that is computed in the initialization step (Talischi et al. 2012).

The remainder of the steps follow an iterative procedure. In the analysis step, the system matrices are obtained using the current values for the design variables, and then, the covariance of the response is computed by solving the Lyapunov equation. In the sensitivity step, the adjoint Lyapunov equation is solved to obtain the Lagrange multiplier and the performance function sensitivity; the constraint sensitivity is computed directly. In the update step, the new values for the design variables are obtained by using the method of moving asymptotes (Svanberg 1987; 2002). The iterative scheme is applied until the maximum change in the design variables is below a specified threshold.



Fig. 4 Topology optimization flowchart for structures with stochastic dynamic loads

4 Numerical examples

In this section, the proposed framework is illustrated through four examples: (i) minimization of the displacement of a mass at the free end of a cantilever beam subjected to a stochastic dynamic base excitation, (ii) minimization of tip displacement of a cantilever beam subjected to a stochastic dynamic tip load, (iii) minimization of tip displacement and acceleration of a cantilever beam subjected to a stochastic dynamic tip load, and (iv) minimization of a plate subjected to multiple stochastic dynamic loads.

4.1 Displacement minimization of a cantilever beam with stochastic base excitation

This example minimizes the displacement of a point mass placed at the free end of the based-excited cantilever beam that was recently considered by Yang et al. (2017). The following parameters are taken from their study. The design domain for the cantilever beam is given by a 10 m \times 20 m rectangle, which is shown in Fig. 5. The solid linear elastic material has the following properties: Young's modulus $E^0 = 10$ kPa, Poisson's ratio $\nu = 0.3$, density $\rho^0 = 1 \frac{\text{kg}}{\text{m}^3}$, tip mass m_0 kg, and Ersatz parameter $\epsilon = 10^{-4}$. The domain has a uniform thickness of 1 m, and it is assumed to be in plane stress condition. The continuum domain is discretized using 80×160 Q4 elements. The radius of the filter is equal to 0.25. The volume of the structure is constrained to be less or equal than 0.50 of the solid domain. The damping matrix is obtained using Rayleigh damping with 5% damping ratio for the first and fourth modes. The stochastic base motion $z_g(t)$ is modeled using a secondorder Kanai-Tajimi spectrum with $\omega_g = 8 \text{ rad/s}, \zeta_g = 0.64$, and $S_0 = 0.019$. Note that due to the large damping ratio, this excitation behaves similar to a band-limited white noise with cutoff frequency around 2 Hz.

Topology optimization is performed to minimize the variance of the vertical displacement of the point tip mass $\mathbb{E}(u_{tip}^2) = \sigma_{tip}^2$. Figure 6a shows the optimal design for this example with $m_0 = 20$ kg (i.e., 20% of the allowable



Fig. 5 Rectangular domain geometry for cantilever plate with point mass and dynamic stochastic base excitation



Fig. 6 Optimal topology for base excitation with Kanai-Tajimi spectrum with tip mass $\mathbf{a} m_0 = 20 \text{ kg}$, $\mathbf{b} m_0 = 60 \text{ kg}$, and $\mathbf{c} m_0 = 160 \text{ kg}$; \mathbf{d} static tip load

mass in the solid domain). Note that the optimal topology shown in Fig. 6a differs from the result obtained by Yang et al. (2017), primarily due to approximations employed in their study to facilitate the numerical solution (e.g., using first few modes approximation). An additional difference is that Yang et al. (2017) used a projection filter instead of the filter used in this study. In contrast, the solution in this study readily includes the influence of all the modes. The objective function is equal to 0.073 and the first frequency is equal to 0.265 Hz.

Finally, additional studies are performed for the same domain and excitation but varying the tip mass. Figure 6ac shows the optimal designs for different tip masses, and Fig. 6d shows the optimal design of the domain subjected to a constant static tip load. Figure 6 illustrates that, as the tip mass increases, the optimal topology evolves toward the optimal topology corresponding to a fixed-base beam with a static tip force Moreover, for this example, if the tip mass is larger than six times the allowable mass in the solid domain, then the optimal topology is nearly identical to the static solution. This phenomenon occurs for such large masses, because the problem essentially reduces to a single DOF system. For this example, the excitation appears as a white noise to the structure as the frequencies become smaller with increasing mass; because the displacement covariance of a single DOF system subjected to a white noise is inversely proportional to the natural frequency of the structure raised to 1.5 power, minimization of the tip covariance is equivalent to maximizing the stiffness, which is the goal of static compliance minimization.

4.2 Displacement minimization of a cantilever beam with stochastic dynamic tip load

This next example explores a cantilever beam subjected to a stochastic dynamic load at the center of the free end of the beam as shown in Fig. 7. The objective is to minimize the displacement at the point of application of the force. The design domain for the cantilever beam is given by the $6 \text{ m} \times 12 \text{ m}$ rectangle, which is composed of a solid linear elastic material having the following properties, which are representative of structural steel: Young's modulus E^0 =



Fig. 7 Rectangular domain geometry for cantilever plate with dynamic stochastic tip load



Fig. 8 Optimal topology for static load

210 GPa, Poisson's ratio $\nu = 0.3$, density $\rho^0 = 7500 \frac{kg}{m^3}$, and Ersatz parameter $\epsilon = 10^{-4}$. The domain has a uniform thickness of 0.10 m, and due to its thickness, the continuum domain is assumed to be in plane stress condition. The continuum domain is discretized using 80 × 160 Q4 elements. The radius of the filter is equal to 0.20. The volume of the structure is constrained to be less or equal than 0.30 of the solid domain. The damping matrix is obtained using Rayleigh damping with 2% damping ratio for the first two modes.

To provide a reference of comparison, the optimal topology is first obtained considering the applied tip force f(t) to be a deterministic constant (i.e., a static force). Minimization of the vertical tip displacement leads to the topology shown in Fig. 8, which is termed the reference design. The frequency response function (FRF) from the input excitation to the tip displacement, $H_{u_{tip}f}(\omega)$, is calculated as the discrete Fourier transform of the numerical impulse response function; this FRF is normalized to have a value of 1 at $\omega = 0$ and shown in Fig. 9. The first four peaks in this FRF, corresponding to the natural frequencies of the static design, are equal to 30 Hz, 41 Hz, 55 Hz, and 75 Hz.

Next, the tip force f(t) is modeled as a band-limited white noise (BLWN), by passing a scalar white noise (WN) with power spectral density equal to S_0 through a 8-pole, low-pass elliptic filter. The filter has a cutoff frequency f_c , a peak-to-peak ripple of 0.1 dB, and stop-band attenuation of 100 dB. Subsequently, this filter is cast into the state space representation given in (5).

Topology optimization is then performed to minimize the variance of the vertical tip displacement $\mathbb{E}(u_{tip}^2) = \sigma_{tip}^2$ for the cantilever beam subjected to a dynamic stochastic tip loading. The cutoff frequencies are chosen with respect to the reference structure, instead of the initial uniform design, to make a direct comparison with this optimal structure. As discussed below, when the cutoff frequency is zero (i.e., the stochastic load is just a random variable), then the static optimal design and optimal design for the stochastic load are identical. Four cases are considered as follows: (i) $f_{\rm c} = 7$ Hz, which is below the first natural frequency of the reference design; (ii) $f_c = 35$ Hz, which is between the first and second natural frequencies of the reference domain; (iii) $f_c = 60$ Hz, which is between the second and third natural frequencies of the reference design; and (iv) $f_c = \infty$ Hz, (i.e., the tip force f(t) is a white noise). For case (iv), the augmented state space matrices A_a and B_a in (12) are replaced those for the structural system A_s and \mathbf{B}_{s} . The magnitude of the FRF of the various elliptic filters is superimposed on the normalized FRF of the reference design in Fig. 8.

Figure 10a–d shows the optimal design for the four cases. These figures show that the optimal topology depends on the cutoff frequency. For case (i), the optimal topology matches the reference design. An additional optimization was performed using a cutoff frequency of 15 Hz; these results were also visually indistinguishable from the static design, and are thus not presented here.

To better understand why the optimal topology for a small cutoff frequency matches the reference design in







Fig. 10 Optimal topologies for a BLWN with $f_c = 7$ Hz, b BLWN with $f_c = 35$ Hz, c BLWN with $f_c = 60$ Hz, and d white noise

this example, consider the stationary variance of the tip displacement, which is equal to the area under the curve of the power spectral density (PSD) of the tip response $S_{uu}(\omega)$,

$$\sigma_{\rm tip}^2 = 2 \int_0^\infty S_{uu}(\omega) d\omega \tag{37}$$

The PSD of the response depends on the PSD of the input force and the FRF of the structure $H_{uf}(\omega)$ as follows

$$S_{uu}(\omega) = \left| H_{uf}(\omega) \right|^2 S_{ff}(\omega)$$
(38)

Because f(t) is a BLWN, the PSD of the response can be well-approximated as

$$\sigma_{\rm tip}^2 \approx 2S_0 \int_0^{2\pi f_{\rm c}} \left| H_{uf}(\omega) \right|^2 d\omega \tag{39}$$

If the BLWN cutoff frequency f_c is sufficiently below the first natural frequency of the structure (i.e., the pseudo-static region), then $H_{uf}(\omega)$ is constant (see Fig. 11), then

$$\sigma_{\rm tip}^2 \approx 4\pi f_{\rm c} S_0 \left| H_{uf}(0) \right|^2 \tag{40}$$





which is proportional to the square of the tip displacement due to a constant static load. Therefore, in this example, for stochastic dynamic excitation with frequency content that is well below the first natural frequency, the optimal topology can be expected to coincide with the static topology optimization case.

In contrast, for large cutoff frequencies, i.e., $f_c \rightarrow \infty$, the excitation approaches a white noise, and consequently, the optimal topology coincides with the optimal topology for a pure white noise. In this particular example, the optimization was performed using cutoff frequencies of 300 Hz and 500 Hz; the results are visually indistinguishable from the white noise design and thus are not presented here.

To understand the influence of the cutoff frequency f_c in the optimal topology, the FRF of the different designs are obtained as described previously and their magnitudes are shown in Fig. 11. Note that the first frequency of the static design is represented by a large peak in the magnitude of the FRF; therefore, for $f_c = 35$ Hz, i.e., the cutoff is greater than the first natural frequency, the resulting topology considerably reduces the peak of the first frequency and also produces a shift of the first frequency. However, in this case, the second peak has a magnitude of the same order of the first peak of the static design, but this larger modal response does not increase the covariance because this peak is located in the stop-band of the low-pass filter. Similarly, for $f_c = 60$ Hz, i.e., the cutoff is greater than the third natural frequency, the resulting topology considerably reduces the peak of the first and second frequencies and also produces a shift of these frequencies. However, in this case, the third peak has shifted outside of the bandwidth of the excitation, so the larger magnitude does not increase the covariance. For the white noise excitation, the covariance of the response is the area under the magnitude of the FRF over the entire frequency domain.

Finally, Fig. 12 shows a comparison of the covariance of the tip displacement of the different designs using different excitations; the excitation cutoff frequency is shown on the x-axis, and the normalized covariance of the response is shown on the y-axis. The covariances for each cutoff frequency are normalized by the covariance of the corresponding optimal design. The different bars correspond to the designs described in the legend. As expected, the minimum covariance for each design is obtained by the case designed for the corresponding cutoff frequency. For a small cutoff frequency, the static design and the 7 Hz design perform considerably better than the other designs. For a cutoff frequency between the first and second modes of the static case, the case designed for this frequency performs better than the other designs; meanwhile, the static design yields a very large covariance, and the other dynamic designs perform better. For a cutoff frequency between the second and third modes of the static case, the case



Fig. 12 Normalized covariance of the tip displacement of the different designs using different excitations, where the covariance is normalized by the minimum covariance among the designs

designed for this frequency performs better than the other designs; meanwhile, the first dynamic design yield a very large covariance. For a white noise, the case designed for this excitation performs best, with the other designs having poorer performance. In summary, the best performance is achieved when the excitation can be appropriately modeled in the topology optimization problem formulation.

4.3 Multi-objective minimization of a cantilever beam with stochastic dynamic tip load

Xu et al. (2017a) showed that interstory displacements and accelerations are often competing performance objectives; therefore; in this example, the covariance of the vertical tip displacement and acceleration of a cantilever beam subjected to a dynamic stochastic tip loading are minimized. This task is done by means of a combined performance function given by a weighted average of both responses

$$J = \mathbb{E}(u_{\rm tip}^2) + \alpha \mathbb{E}(\ddot{u}_{\rm tip}^2) \tag{41}$$

where the parameter α determines the influence of acceleration response in the performance function: small values of α correspond to displacement optimization and large values correspond to acceleration optimization. The design domain is the same 6 × 12 rectangle of Section 4.2, which is shown in Fig. 7. As in the previous example, the tip load f(t) is modeled as a band-limited white noise (BLWN).

Topology optimization is performed to minimize the multi-objective function for the following excitation cutoff



Fig. 13 Optimal topologies for BLWN excitation with $f_c = 35$ Hz and multi-objective performance function (41) with $\mathbf{a} \alpha = 10^{-8}$, $\mathbf{b} \alpha = 10^{-7}$, $\mathbf{c} \alpha = 10^{-6}$, and $\mathbf{d} \alpha = 10^{-5}$

frequencies: (i) $f_c = 35$ Hz, which is between the first and second natural frequencies of the reference domain, and (ii) $f_c = 60$ Hz, which is between the second and third natural

frequencies of the reference design. Also, different values of the parameter α are considered for each of the two cutoff frequencies.



Fig. 14 Optimal topologies for BLWN excitation with $f_c = 60$ Hz and multi-objective performance function (41) with $\mathbf{a} \alpha = 10^{-9}$, $\mathbf{b} \alpha = 10^{-8}$, $\mathbf{c} \alpha = 10^{-7}$, and $\mathbf{d} \alpha = 10^{-6}$

The stationary variance of the tip displacement and acceleration are given by the area under the curve of the PSD of the corresponding response; therefore, the objective function is given by

$$J = 2 \int_0^\infty \left(\left| H_{uf}(\omega) \right|^2 + \alpha \left| H_{\ddot{u}f}(\omega) \right|^2 \right) S_{ff}(\omega) d\omega$$

= $2 \int_0^\infty (1 + \alpha \omega^4) \left| H_{uf}(\omega) \right|^2 S_{ff}(\omega) d\omega$ (42)

Because f(t) is a BLWN, the objective function can be well-approximated as

$$J \approx 2S_0 \int_0^{2\pi f_c} (1 + \alpha \omega^4) \left| H_{uf}(\omega) \right|^2 d\omega$$
(43)

For small cutoff frequencies f_c , the acceleration influence in the performance function is small due to the quartic

Fig. 15 Comparison of the FRF of the topologies for different multi-objective performance functions

term in the integral. Therefore, for stochastic dynamic excitation with small cutoff frequency, the optimal topology can be expected to coincide with the displacement-only optimization.

Figure 13a–d shows the optimal design for $f_c = 35$ Hz with increasing values of α . These figures show that for this fixed cutoff frequency, the optimal topology depends on the parameter α . As seen here, when the weighting parameter is small, the optimal topology approximates the displacement only optimization $\alpha = 0$, which is shown in Fig. 10b, and for larger values of α , the topology is modified considerably, with more mass being allocated to the top part of the structure.

Figure 14a–d shows the optimal design for $f_c = 60$ Hz with increasing values of α . These figures show that for this fixed cutoff frequency, the optimal topology depends



(a) BLWN with $f_c = 35$ Hz, cutoff frequency is shown as the dashed line



(b) BLWN with $f_{\rm c} = 60$ Hz, cutoff frequency is shown as the dashed line





on the parameter α . As also seen here, when the weighting parameter is small, the optimal topology approximates the

displacement only optimization $\alpha = 0$, which is shown in Fig. 10c, and for larger values of α , the topology is modified

considerably, with more mass being allocated in the top part of the structure.

In both cases, the evolution of the optimal topologies can be explained by the quartic term corresponding to the acceleration response in the performance function in (43), which causes the magnitude of the acceleration FRF to increase faster than the displacement counterpart as the frequency increases (i.e., the higher modes are more important to the acceleration response, as compared to displacement response). Figure 15a–b shows the magnitudes of the FRF of the tip displacement to the excitation for all optimal designs obtained for both cutoff frequencies and different values of α in the performance function.

Note that as α increases, the magnitude of the FRF at 0 Hz increases, indicating that the static compliance of the structure increases with α , and the first modal frequency of the structure decreases with increasing α . For the case $f_c = 35$ Hz, the static tip deflection of the optimal topology is increased approximately 8.4 times when α is increased from 0 to 10^{-5} , which indicates that the overall system is more flexible to effectively reduce accelerations but increasing displacements. For completeness, Fig. 16a–h shows the first and second mode shapes for the optimal topologies in Figs. 13a, d and 14a, d.

Finally, the Pareto optimal front is calculated to illustrate the tradeoff in the response of displacements and accelerations and plotted in Fig. 17a–b for cutoff frequencies $f_c = 35$ Hz and $f_c = 60$ Hz, respectively. This curve is a plot of the normalized standard deviation of the tip displacement versus the normalized standard deviation of the tip acceleration for different values of α . Here, the standard deviations of the tip displacements and accelerations are normalized by dividing by their



Fig. 18 Rectangular domain geometry for a plate with multiple dynamic stochastic loads

corresponding minimum. The numbers shown next to each point correspond to the values of α , and note that additional values of α , whose topologies are not shown due to space limitations, are included in these curves. As expected, for small values of α , the displacement response is small but the acceleration response is large, and as the parameter α increases, the displacement response increases and the acceleration response decreases. Points below this Pareto optimal front correspond to infeasible designs and above this curve correspond to non-optimal designs. Also, the displacement response is more affected for the first cutoff frequency than for the second cutoff frequency, and the acceleration response is more affected for the second cutoff frequency than in the first cutoff frequency. This difference



Fig. 17 Normalized standard deviation of displacement versus normalized standard deviation of acceleration curve for optimal designs with different objective functions (41)



Fig. 19 Optimal topologies for a static loads and b multiple BLWN excitations with correlation coefficient $\rho = 0.5$

can be explained by the fact that the acceleration response is more dependent on higher frequencies. In summary, as the weighting parameter increases, the acceleration response is more important in the objective function, which modifies substantially the optimal topology from the displacementonly optimization; and as the cutoff frequency increases, the acceleration response becomes more important.

4.4 Minimization of a plate with multiple stochastic dynamic loads

This example demonstrates the suitability of the proposed methodology for multiple-input multiple-output systems. The domain and loads are shown in Fig. 18, where f_1 and f_2 are different stochastic processes applied on points 1 and 2, respectively. The design domain for the cantilever beam is given by the 5 m \times 5 m rectangle, which is composed of a solid linear elastic material having the following properties, which are representative of structural steel: Young's modulus $E^0 = 210$ GPa, Poisson's ratio $\nu = 0.3$, density $\rho^0 = 7500 \frac{kg}{m^3}$, and Ersatz parameter $\epsilon = 10^{-4}$. The domain has a uniform thickness of 0.10 m, and due to its thickness, the continuum domain is assumed to be in plane stress condition. The continuum domain is discretized using 100×100 Q4 elements. The radius of the filter is equal to 0.20. The volume of the structure is constrained to be less or equal than 0.30 of the solid domain. The damping matrix is obtained using Rayleigh damping with 2% damping ratio for the first two modes. The objective function is given by

$$J = \mathbb{E}(u_1^2) + \mathbb{E}(u_2^2) \tag{44}$$

where u_1 and u_2 are the lateral and vertical displacements of points 1 and 2.

To provide a reference of comparison, the optimal topology is first obtained considering the applied forces to be deterministic constants, i.e., $f_1(t) = f_2(t) = P$ where P is a deterministic constant. Minimization of the sum of displacements leads to the topology shown in Fig. 19a, which is termed the reference design.

The stochastic processes are modeled as correlated bandlimited white noises, so that the excitation matrices are composed of independent scalar low-pass elliptic filters arranged in diagonal blocks. The cutoff frequency of both scalar filters is equal to 50 Hz. The magnitude two-sided constant power spectral density matrix is given by

$$\mathbf{S}_0 = \begin{bmatrix} 1 & \rho \\ \rho & 1 \end{bmatrix} \tag{45}$$

where ρ is the correlation between the two loading processes. Note that $|\rho| < 1$ to assure the positive definiteness of the previous matrix. If $\rho = 1$, then both processes are the same, and it is treated as a single-input system. If $\rho = -1$, the system is treated as a single-input, and the processes are inversely correlated, i.e., in Fig. 18, one of the loading arrows is reversed.

Figure 19b shows the optimal design for multiple stochastic dynamic loads with correlation coefficient $\rho = 0.5$. These figures demonstrate that for this fixed cutoff frequency, the optimal topology varies considerably when considering the forces to be stochastic.

5 Conclusions

This paper has proposed a framework for topology optimization of stochastically excited structures. The input

was modeled as a filtered white noise, and the performance of the structure due to this excitation was given in terms of the covariance of the stationary structural responses. The objective function for the optimization was defined as the trace of the product of a positive semidefinite symmetric matrix and the covariance of the stationary response. The covariances were obtained by solving a large-scale Lyapunov equation using an algorithm which is efficient both in terms of memory and computational time. The objective function was shown to be general enough to represent displacement, interstory drifts, velocities, and accelerations at one or many points, making it suitable for multi-objective optimization. A volume constraint was imposed to limit the design space, and the design variables were chosen as the relative densities in each element, which were bounded to achieve physically meaningful solutions. The material properties for intermediate densities were obtained using the SIMP interpolation rule; a linear hat filter was used to avoid numerical instabilities. An efficient adjoint method to obtain the sensitivities of the performance function was proposed, which requires the solution of an adjoint Lyapunov equation, also solved using the efficient Lyapunov equation solver. Iterations were carried out using a gradient-based approach commonly employed in the topology optimization field.

The proposed framework was illustrated by conducting topology optimization of cantilever beams with different excitations and performance functions. The first example considers displacement optimization of a mass on the end of a cantilever beam with stochastic dynamic base motion. The topology obtained using the proposed approach differs from the result found in a previous study, primarily due to approximations employed in that study to achieve numerical solutions of the topology optimization problem. In the proposed approach, these approximations were not required. Local mode penalization improves the numerical behavior eliminating disconnected regions. Also, as the tip mass increases, the optimal topology evolves towards the solution obtained for a vertical static tip load.

The second example minimized the tip displacement for a cantilever beam subjected to a stochastic dynamic point force. The force was modeled as a band-limited white noise (BLWN) using an 8-pole elliptic filter with different cutoff frequencies and as a pure white noise. First, optimization was performed to minimize the covariance of the tip displacement. For cutoff frequencies well below the first natural frequency, the optimal topology coincides with the static case. For large cutoff frequencies, the excitation approaches a white noise, and consequently, the optimal topology coincides with the optimal topology obtained for a pure white noise excitation. In this example, for frequencies above 300 Hz, the optimal topology for the BLWN and the white noise was indistinguishable. The static design is outperformed by the dynamic designs when the cutoff frequency is larger than its first frequency. In general, the best performance is achieved when the excitation can be appropriately modeled in the topology optimization problem formulation.

Also, the proposed framework was employed to demonstrate the tradeoffs between minimizing tip displacement and acceleration. The cantilever beam subjected to a dynamic stochastic tip force was again examined, where the force was modeled as a band-limited white noise. For a fixed cutoff frequency, the optimal topology depends on the weighting parameter, α . For small values of α , the optimal topology is close to the displacement-only optimization, and for larger values of α , the topology is modified considerably, with more mass being allocated in the top part of the structure. The evolution of the optimal topologies with increasing values of α is due to the magnitude of the acceleration FRF that increases faster than the displacement counterpart as the frequency increases. Moreover, the static compliance increases and the first modal frequency of the structure decreases with increasing α , both of which indicate reductions in the acceleration are obtained by making the overall system more flexible. In addition, the Pareto optimal curve is introduced to explore the tradeoffs between reducing displacements and accelerations. For small values of α , the displacement response is small but the acceleration response is large, and as the parameter α increases, the displacement response increases and the acceleration response decreases. Also, as the cutoff frequency increases, the acceleration response becomes more important.

The results presented herein demonstrate the efficacy of the proposed approach for multi-objective topology optimization of stochastically excited structures. This framework can also accommodate multi-objective problems as well as multiple input-multiple output systems.

Appendix: Derivations

Performance functions

The details on how to obtain the matrix **F** for the performance functions described in Section 2.4 are described in this section. Initial notation is used for simplicity, the indices take values from 1, 2, ..., N unless noted otherwise, and the subscript of the covariance of the response of the augmented system is dropped for simplicity. 1) Covariance of the various DOFs defined by the output equation $\mathbf{y} = \mathbf{C}_{a}\mathbf{x}_{a}$, therefore $\mathbf{\Gamma}_{y} = \mathbf{C}_{a}\mathbf{\Gamma}_{\mathbf{x}_{a}}\mathbf{C}_{a}^{T}$

$$J = \mathbf{C}_{a} \mathbf{\Gamma}_{\mathbf{x}_{a}} \mathbf{C}_{a}^{\mathrm{T}} = C_{ki} \Gamma_{ij} C_{kj} = C_{ki} C_{kj} \Gamma_{ij}$$
$$= (C^{\mathrm{T}} C)_{ij} \Gamma_{ij} = (\mathbf{C}_{a}^{\mathrm{T}} \mathbf{C}_{a}) : \mathbf{\Gamma}_{\mathbf{x}_{a}}$$
(46)

$$\implies \mathbf{F} = (\mathbf{C}_{\mathbf{a}}^{\mathrm{T}} \mathbf{C}_{\mathbf{a}}) \tag{47}$$

F is clearly symmetric positive semi-definite.

The following shows examples of performance functions

$$J = \mathbb{E}(u_p^2) \implies \mathbf{C}_{\mathbf{a}} = (C_u)_{p,:}$$

$$J = \mathbb{E}((u_p - u_q)^2) \implies \mathbf{C}_{\mathbf{a}} = (C_u)_{p,:} - (C_u)_{q,:}$$

$$J = \mathbb{E}(\sum_p u_p^2) \implies \mathbf{C}_{\mathbf{a}} = \sum_p (C_u)_{p,:}$$

$$J = \mathbb{E}(\dot{u}_p^2) \implies \mathbf{C}_{\mathbf{a}} = (C_{\dot{u}})_{p,:}$$

$$J = \mathbb{E}((\ddot{u}_p^{\text{abs}})^2) \implies \mathbf{C}_{\mathbf{a}} = (C_{\ddot{u}})_{p,:}$$
(48)

where $X_{p,:}$ denotes the *p*th row of matrix **X** expected static compliance

$$J = \mathbb{E}(\mathbf{u}^{T}\mathbf{K}\mathbf{u}) = \mathbb{E}(u_{i}K_{ij}u_{j}) = K_{ij}\mathbb{E}(u_{i}u_{j})$$
$$= K_{ij}(\Gamma_{u})_{ij} = \mathbf{K}: \Gamma_{u}$$
(49)

where $\Gamma_u = \mathbb{E}(\mathbf{u}\mathbf{u}^T)$ is the covariance of the displacement, then $\Gamma_u = \mathbf{C}_{\mathbf{u}}\Gamma_{\mathbf{x}a}\mathbf{C}_{\mathbf{u}}^T$, and consequently

$$J = K_{ij}(C_u \Gamma C_u^{\mathrm{T}})_{ij} = K_{ij}(C_u)_{ip}(\Gamma)_{pq}(C_u)_{jq}$$

= $(C_u^{\mathrm{T}} K C_u)_{pq}(\Gamma)_{pq} = (\mathbf{C}_{\mathbf{u}}^{\mathrm{T}} \mathbf{K} \mathbf{C}_{\mathbf{u}}) : \mathbf{\Gamma}_{\mathbf{x}_{\mathrm{a}}}$ (50)

$$\implies \mathbf{F} = \mathbf{C}_{\mathbf{u}}^{\mathrm{T}} \mathbf{K} \mathbf{C}_{\mathbf{u}} \tag{51}$$

Because \mathbf{K} is symmetric positive definite, \mathbf{F} is clearly symmetric positive semidefinite.

3) expected kinetic energy

2)

$$J = \mathbb{E}(\dot{\mathbf{u}}^{\mathrm{T}}\mathbf{M}\dot{\mathbf{u}}) = \mathbb{E}(\dot{u}_{i}M_{ij}\dot{u}_{j}) = M_{ij}\mathbb{E}(\dot{u}_{i}\dot{u}_{j})$$
$$= M_{ij}(\Gamma_{\dot{u}})_{ij} = \mathbf{M}: \Gamma_{\dot{\mathbf{u}}}$$
(52)

where $\Gamma_{\dot{u}} = \mathbb{E}(\dot{u}\dot{u}^{T})$ is the covariance of the velocity, then $\Gamma_{\dot{u}} = C_{\dot{u}}\Gamma_{x_{a}}C_{\dot{u}}^{T}$, and consequently

$$J = M_{ij}(C_{\dot{u}}\Gamma C_{\dot{u}}^{\mathrm{T}})_{ij} = M_{ij}(C_{\dot{u}})_{ip}(\Gamma)_{pq}(C_{\dot{u}})_{jq} = (C_{\dot{u}}^{\mathrm{T}}MC_{\dot{u}})_{pq}(\Gamma)_{pq} = (\mathbf{C}_{\dot{u}}^{\mathrm{T}}\mathbf{M}\mathbf{C}_{\mathbf{B}}) : \mathbf{\Gamma}_{\mathbf{x}_{a}}$$
(53)

$$\implies \mathbf{F} = \mathbf{C}_{\dot{\mathbf{u}}}^{\mathrm{T}} \mathbf{M} \mathbf{C}_{\mathbf{F}}$$
(54)

Because \mathbf{M} is symmetric positive semidefinite, \mathbf{F} is clearly symmetric positive semidefinite.

 linear combination of previous performance functions with non-negative coefficients, i.e., the performance function is given by

$$J = \alpha J_1 + \beta J_2 \tag{55}$$

where $J_1 = \mathbf{F}_1 : \mathbf{\Gamma}_{\mathbf{x}_a}$ and $J_2 = \mathbf{F}_2 : \mathbf{\Gamma}_{\mathbf{x}_a}$, and α and β are non-negative real numbers. Then,

$$J = \alpha \mathbf{F}_1 : \mathbf{\Gamma}_{\mathbf{x}_a} + \beta \mathbf{F}_2 : \mathbf{\Gamma}_{\mathbf{x}_a} = (\alpha \mathbf{F}_1 + \beta \mathbf{F}_2) : \mathbf{\Gamma}_{\mathbf{x}_a}$$
(56)

$$\implies \mathbf{F} = \alpha \mathbf{F}_1 + \beta \mathbf{F}_2 \tag{57}$$

Because in this case \mathbf{F} is a linear combination of \mathbf{F} in the previous cases with non-negative coefficients, it is clearly symmetric positive semidefinite.

Sensitivity analysis

The details on how to obtain the sensitivity of the performance functions described in Section 3.3 are described in this section. Initial notation is used and the subscript of the covariance of the response of the augmented system is dropped for simplicity, and the indices take values from $1, 2, \ldots, N'$ unless noted otherwise. The sensitivity of the performance function is given by

$$\frac{\partial J}{\partial z_n} = \frac{\partial \phi}{\partial z_n} + \frac{\partial \phi}{\partial \Gamma_{ij}} \frac{\partial \Gamma_{ij}}{\partial z_n}$$
(58)

which requires the sensitivity of covariance matrix that is implicitly defined by (21). Differentiation to this equation yields the following equations

$$\mathbf{A}_{a}\frac{\partial \mathbf{\Gamma}}{\partial z_{n}} + \frac{\partial \mathbf{\Gamma}}{\partial z_{n}}\mathbf{A}_{a}^{\mathrm{T}} + \mathbf{Q}_{n} = \mathbf{0} \quad \text{where} \\ \mathbf{Q}_{n} = \frac{\partial \mathbf{A}_{a}}{\partial z_{n}}\mathbf{\Gamma} + \mathbf{\Gamma}\frac{\partial \mathbf{A}_{a}^{\mathrm{T}}}{\partial z_{n}} + \frac{\partial (\tilde{\mathbf{B}}_{a}\tilde{\mathbf{B}}_{a}^{\mathrm{T}})}{\partial z_{n}}$$
(59)

This means that the direct differentiation method requires the solution of one Lyapunov equation for each element, which makes the overall process a quartic order process. Note also, that the \mathbf{Q}_n is not necessarily a low-rank matrix product, and consequently, the more efficient method described in Section 3.2 may not be applied.

An adjoint method is applied to make the process more efficient, in terms of the Lagrangian function expressed in (27), which is rewritten and reordered next in initial notation

$$\mathcal{L} = J + \Lambda_{ij} (A_{a,ik} \Gamma_{kj} + \Gamma_{ik} A_{a,jk} + Q_{ij})$$

= $J + (A_{a,ki} \Lambda_{kj} + \Lambda_{ik} A_{a,kj}) \Gamma_{ij} + \Lambda_{ij} Q_{ij}$ (60)

where $\mathbf{Q} = \tilde{\mathbf{B}}_{a} \tilde{\mathbf{B}}_{a}^{T}$. Differentiating the previous equation yields the following

$$\frac{\partial \mathcal{L}}{\partial z_n} = \frac{\partial J}{\partial z_n} + \frac{\partial (A_{a,ki}\Lambda_{kj} + \Lambda_{ik}A_{a,kj})}{\partial z_n}\Gamma_{ij} + (A_{a,ki}\Lambda_{kj} + \Lambda_{ik}A_{a,kj})\frac{\partial\Gamma_{ij}}{\partial z_n} + \Lambda_{ij}\frac{\partial Q_{ij}}{\partial z_n}$$
(61)

Plugging (58) in the previous equation and reordering yields

$$\frac{\partial \mathcal{L}}{\partial z_n} = \frac{\partial (A_{\mathbf{a},ki}\Lambda_{kj} + \Lambda_{ik}A_{\mathbf{a},kj})}{\partial z_n}\Gamma_{ij} + \Lambda_{ij}\frac{\partial Q_{ij}}{\partial z_n} + \left(A_{\mathbf{a},ki}\Lambda_{kj} + \Lambda_{ik}A_{\mathbf{a},kj} + \frac{\partial \phi}{\partial \Gamma_{ij}}\right)\frac{\partial \Gamma_{ij}}{\partial z_n} + \frac{\partial \phi}{\partial z_n}$$
(62)

To remove the dependence on the implicit derivative of the covariance matrix, the first factor in the third term in the RHS of the previous equation is defined as 0, that is

$$A_{a,ki}\Lambda_{kj} + \Lambda_{ik}A_{a,kj} + \frac{\partial\phi}{\partial\Gamma_{ij}} = 0$$
(63)

which written in abstract form yields (28). Therefore, the gradient of the performance function is given by

$$\frac{\partial J}{\partial z_n} = \left(\frac{\partial A_{\mathbf{a},ik}}{\partial z_n}\Gamma_{kj} + \Gamma_{ik}\frac{\partial A_{\mathbf{a},jk}}{\partial z_n} + \frac{\partial Q_{ij}}{\partial z_n}\right)\Lambda_{ij} + \frac{\partial \phi}{\partial z_n}$$
(64)

and the previous equation written in abstract form gives (29).

Then, the sensitivity of the performance function requires the derivatives of the matrices A_a and B_a

$$\frac{\partial \mathbf{A}_{a}}{\partial z_{n}} = \begin{bmatrix} \frac{\partial \mathbf{A}_{s}}{\partial z_{n}} & \frac{\partial \mathbf{B}_{s}}{\partial z_{n}} \mathbf{C}_{f} \\ \mathbf{0}_{N_{f} \times 2N} & \mathbf{0}_{N_{f} \times N_{f}} \end{bmatrix}, \quad \frac{\partial \mathbf{B}_{a}}{\partial z_{n}} = \mathbf{0}$$
(65)

The derivative of state matrices is given by

$$\frac{\partial \mathbf{A}_{s}}{\partial z_{n}} = \begin{bmatrix} \mathbf{0}_{N \times N} & \mathbf{0}_{N \times N} \\ \frac{\partial (-\mathbf{M}^{-1}\mathbf{K})}{\partial z_{n}} & \frac{\partial (-\mathbf{M}^{-1}\mathbf{C})}{\partial z_{n}} \end{bmatrix}$$
(66)

$$\frac{\partial \mathbf{B}_{s}}{\partial z_{n}} = \begin{bmatrix} \mathbf{0}_{N \times 1} \\ \frac{\partial (\mathbf{M}^{-1} \mathbf{G})}{\partial z_{n}} \end{bmatrix}$$
(67)

The matrices \mathbf{M} and \mathbf{K} are obtained using an assembly process described by

$$\mathbf{M}(\mathbf{z}) = \mathop{\mathbf{A}}_{n=1}^{N_{\text{el}}} \rho(z_n) \mathbf{M}_n^0, \quad \mathbf{K}(\mathbf{z}) = \mathop{\mathbf{A}}_{n=1}^{N_{\text{el}}} E(z_n) \mathbf{K}_n^0$$
(68)

where \mathbf{M}_n^0 and \mathbf{K}_n^0 are the mass and stiffness matrices of element *n* with solid material. The derivatives of them are given by

$$\frac{\partial \mathbf{M}}{\partial z_n} = \frac{\partial \rho}{\partial z}(z_n)\mathbf{M}_n^0, \quad \frac{\partial \mathbf{K}}{\partial z_n} = \frac{\partial E}{\partial z}(z_n)\mathbf{K}_n^0$$
(69)

The derivative of M^{-1} is

$$\frac{\partial (\mathbf{M}^{-1})}{\partial z_n} = -\mathbf{M}^{-1} \frac{\partial \mathbf{M}}{\partial z_n} \mathbf{M}^{-1}$$
(70)

which yields the following derivatives that appear in the state matrices

$$\frac{\partial (-\mathbf{M}^{-1}\mathbf{K})}{\partial z_n} = \mathbf{M}^{-1} \left(\frac{\partial \mathbf{M}}{\partial z_n} \mathbf{M}^{-1} \mathbf{K} - \frac{\partial \mathbf{K}}{\partial z_n} \right)$$

$$\frac{\partial (-\mathbf{M}^{-1}\mathbf{C})}{\partial z_n} = \mathbf{M}^{-1} \left(\frac{\partial \mathbf{M}}{\partial z_n} \mathbf{M}^{-1} \mathbf{C} - \frac{\partial \mathbf{C}}{\partial z_n} \right)$$

$$\frac{\partial (\mathbf{M}^{-1}\mathbf{G})}{\partial z_n} = \mathbf{M}^{-1} \left(-\frac{\partial \mathbf{M}}{\partial z_n} \mathbf{M}^{-1} \mathbf{G} + \frac{\partial \mathbf{G}}{\partial z_n} \right)$$
(71)

If Rayleigh damping is used, the damping matrix is equal to

$$\mathbf{C} = \alpha_1 \mathbf{M} + \alpha_2 \mathbf{K} \tag{72}$$

and consequently, the following derivative is given by

$$\frac{\partial (-\mathbf{M}^{-1}\mathbf{C})}{\partial z_n} = \alpha_2 \mathbf{M}^{-1} \left(\frac{\partial \mathbf{M}}{\partial z_n} \mathbf{M}^{-1} \mathbf{K} - \frac{\partial \mathbf{K}}{\partial z_n} \right)$$
(73)

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

References

- Allahdadian S, Boroomand B (2016) Topology optimization of planar frames under seismic loads induced by actual and artificial earthquake records. Eng Struct 115:140–154. https://doi.org/10.1016/j. engstruct.2016.02.022, http://www.sciencedirect.com/science/ article/pii/S0141029616001103
- Balling RJ, Balling LJ, Richards PW (2009) Design of Buckling-Restrained braced frames using nonlinear time history analysis and optimization. J Struct Eng 135(5):461–468. https://doi.org/10. 1061/(ASCE)ST.1943-541X.0000007
- Balzer LA (1980) Accelerated convergence of the matrix sign function method of solving Lyapunov, Riccati and other matrix equations. Int J Control 32(6):1057–1078. https://doi.org/10.1080/ 00207178008910040
- Beghini LL, Beghini A, Katz N, Baker WF, Paulino GH (2014) Connecting architecture and engineering through structural topology optimization. Eng Struct 59:716–726. https://doi.org/10.1016/j. engstruct.2013.10.032, http://www.sciencedirect.com/science/ article/pii/S0141029613005014
- Behrou R, Guest JK (2017) Topology optimization for transient response of structures subjected to dynamic loads. In: 18th AIAA/ISSMO Multidisciplinary Analysis and Optimization Conference American Institute of Aeronautics and Astronautics, Reston. https://doi.org/10.2514/6.2017-3657
- Bendsøe MP, Kikuchi N (1988) Generating optimal topologies in structural design using a homogenization method. Comput Methods Appl Mech Eng 71(2):197–224. https://doi.org/10.1016/0045-7825(88)90086-2
- Bendsøe MP, Sigmund O (1999) Material interpolation schemes in topology optimization. Arch Appl Mech (Ingenieur Archiv) 69(9-10):635–654. https://doi.org/10.1007/s004190050248
- Bendsøe MP, Sigmund O (2003) Topology optimization: theory, methods, and applications. Springer, Berlin
- Benner P, Li JR, Penzl T (2008) Numerical solution of largescale Lyapunov equations, Riccati equations, and linear-quadratic optimal control problems. Numer Linear Algebra Appl 15(9):755– 777. https://doi.org/10.1002/nla.622
- Chun J, Song J, Paulino GH (2016) Structural topology optimization under constraints on instantaneous failure probability. Struct Multidiscip Optim 53(4):773–799. https://doi.org/10.1007/s00158-015-1296-y
- Díaz A, Sigmund O (1995) Checkerboard patterns in layout optimization. Struct Optim 10(1):40–45. https://doi.org/10.1007/ BF01743693
- Du J, Olhoff N (2007) Topological design of freely vibrating continuum structures for maximum values of simple and multiple eigenfrequencies and frequency gaps. Struct Multidiscip Optim 34(2):91–110. https://doi.org/10.1007/s00158-007-0101-y
- Fidkowski K, Kroo I, Willcox K, Engelson F (2008) Stochastic Gust Analysis Techniques for Aircraft Conceptual Design. In:

12th AIAA/ISSMO Multidisciplinary Analysis and Optimization Conference American Institute of Aeronautics and Astronautics, Reston. https://doi.org/10.2514/6.2008-5848

- Filipov ET, Chun J, Paulino GH, Song J (2016) Polygonal multiresolution topology optimization (polyMTOP) for structural dynamics. Struct Multidiscip Optim 53(4):673–694. https://doi.org/10.1007/ s00158-015-1309-x
- Golub G, Nash S, Van Loan C (1979) A Hessenberg-Schur method for the problem AX + XB= C. IEEE Trans Autom Control 24(6):909–913. https://doi.org/10.1109/TAC.1979.1102170, http://ieeexplore.ieee.org/document/1102170/
- Gomez F, Spencer BF (2017) Topology Optimization of Structures subjected to Stochastic Dynamic Loading. In: ICSSS17, http:// asem17.org/Keynote/k0501A.pdf
- Haftka RT, Adelman HM (1989) Recent developments in structural sensitivity analysis. Struct Optim 1(3):137–151. https://doi.org/10.1007/BF01637334
- Higham NJ (2008) Functions of matrices : theory and computation. Society for Industrial and Applied Mathematics (SIAM, 3600 Market Street, Floor 6, Philadelphia 19104)
- Hu Z, Ma H, Su C (2016) Topology optimization of structures subjected to non-stationary random excitations. In: ISSRI
- Jbilou K, Riquet A (2006) Projection methods for large Lyapunov matrix equations. Linear Algebra Appl 415(2-3):344– 358. https://doi.org/10.1016/j.laa.2004.11.004, http://linkinghub. elsevier.com/retrieve/pii/S0024379504004707
- Kang BS, Park GJ, Arora JS (2006) A review of optimization of structures subjected to transient loads. Struct Multidiscip Optim 31(2):81–95. https://doi.org/10.1007/s00158-005-0575-4
- Kohn RV, Strang G (1986) Optimal design and relaxation of variational problems, I. Commun Pur Appl Math 39(1):113–137. https://doi.org/10.1002/cpa.3160390107
- Kressner D (2008) Memory-efficient Krylov subspace techniques for solving large-scale Lyapunov equations. In: 2008 IEEE International Conference on Computer-Aided Control Systems. IEEE, pp 613–618. https://doi.org/10.1109/CACSD.2008.4627370, http:// ieeexplore.ieee.org/document/4627370/
- Li JR, White J (2002) Low rank solution of lyapunov equations. SIAM J Matrix Anal Appl 24(1):260–280. https://doi.org/10.1137/ S0895479801384937
- Li J, Chen J (2009) Stochastic dynamics of structures. wiley, Chichester. https://doi.org/10.1002/9780470824269
- Ma ZD, Kikuchi N, Cheng HC (1995) Topological design for vibrating structures. Comput Methods Appl Mech Eng 121(1-4):259–280. https://doi.org/10.1016/0045-7825(94)00714-X, http://linkinghub.elsevier.com/retrieve/pii/004578259400714X
- Olhoff N (1976) Optimization of vibrating beams with respect to higher order natural frequencies. J Struct Mech 4(1):87–122. https://doi.org/10.1080/03601217608907283
- Olhoff N (1989) Multicriterion structural optimization via bound formulation and mathematical programming. Struct Optim 1(1):11– 17. https://doi.org/10.1007/BF01743805
- Penzl T (1999) A cyclic Low-Rank smith method for large sparse lyapunov equations. SIAM J Sci Comput 21(4):1401–1418. https://doi.org/10.1137/S1064827598347666
- Saad Y (1990) Numerical Solution of Large Lyapunov Equations. Signal Process Scattering Oper Theory Numer Methods, Proc MTNS-89 3:503–511. http://citeseerx.ist.psu.edu/viewdoc/ summary?doi=10.1.1.32.2738
- Sigmund O, Petersson J (1998) Numerical instabilities in topology optimization: a survey on procedures dealing with checkerboards, mesh-dependencies and local minima. Struct Optim 16(1):68–75. https://doi.org/10.1007/BF01214002
- Sigmund O (2007) Morphology-based black and white filters for topology optimization. Struct Multidiscip Optim 33(4-5):401– 424. https://doi.org/10.1007/s00158-006-0087-x

- Soong TT, Grigoriu M (1993) Random vibration of mechanical and structural systems. PTR Prentice Hall, https://books.google.com/ books/about/Random_vibration_of_mechanical_and_struc.html? id=6JVRAAAAMAAJ
- Spencer BF, Gomez F, Xu J (2016) Topology optimization for stochastically excited structures. In: ISSRI
- Strang G (2003) Introduction to linear algebra. Wellesley-Cambridge Press, Cambridge
- Svanberg K (1987) The method of moving asymptotes—a new method for structural optimization. Int J Numer Methods Eng 24(2):359– 373. https://doi.org/10.1002/nme.1620240207
- Svanberg K (2002) A class of globally convergent optimization methods based on conservative convex separable approximations. SIAM J Optim 12(2):555–573. https://doi.org/10.1137/ S1052623499362822
- Talischi C, Paulino GH, Pereira A, Menezes IFM (2012) Polytop: a Matlab implementation of a general topology optimization framework using unstructured polygonal finite element meshes. Struct Multidiscip Optim 45(3):329–357. https://doi.org/10.1007/ s00158-011-0696-x
- Tcherniak D (2002) Topology optimization of resonating structures using SIMP method. Int J Numer Methods Eng 54(11):1605– 1622. https://doi.org/10.1002/nme.484
- Wachspress EL (2013) The ADI model problem. Springer, Berlin
- Xu J, Spencer BF, Lu X (2017a) Performance-based optimization of nonlinear structures subject to stochastic dynamic loading. Eng Struct 134:334–345. https://doi.org/10.1016/j.engstruct.2016.12. 051, http://www.sciencedirect.com/science/article/pii/S014102961 6317084
- Xu J, Spencer BF, Lu X, Chen X, Lu L (2017b) Optimization of structures subject to stochastic dynamic loading. Comput-Aided Civ Infrastruct Eng 32(8):657–673. https://doi.org/10.1111/mice. 12274
- Yan K, Cheng G, Wang BP (2016) Adjoint methods of sensitivity analysis for Lyapunov equation. Struct Multidiscip Optim 53(2):225–237. https://doi.org/10.1007/s00158-015-1323-z
- Yang Y, Zhu M, Shields MD, Guest JK (2017) Topology optimization of continuum structures subjected to filtered white noise stochastic excitations. Comput Methods Appl Mech Eng 324:438– 456. https://doi.org/10.1016/J.CMA.2017.06.015, https://www. sciencedirect.com/science/article/pii/S0045782516306648
- Zhang WH, Liu H, Gao T (2015) Topology optimization of large-scale structures subjected to stationary random excitation: An efficient optimization procedure integrating pseudo excitation method and mode acceleration method. Comput Struct 158(C):61–70. https://doi.org/10.1016/j.compstruc.2015.05.027, http://linkinghub.elsevier.com/retrieve/pii/S004579491500173X
- Zhao J, Wang C (2016) Dynamic response topology optimization in the time domain using model reduction method. Struct Multidiscip Optim 53(1):101–114. https://doi.org/10.1007/s00158-015-1328-7
- Zhu JH, Beckers P, Zhang WH (2010) On the multi-component layout design with inertial force. J Comput Appl Math 234(7):2222– 2230. https://doi.org/10.1016/J.CAM.2009.08.073, https://www. sciencedirect.com/science/article/pii/S0377042709005585
- Zhu M, Yang Y, Guest JK, Shields MD (2017) Topology optimization for linear stationary stochastic dynamics: Applications to frame structures. Struct Saf 67:116–131. https://doi.org/10.1016/J. STRUSAFE.2017.04.004, http://www.sciencedirect.com/science/ article/pii/S0167473017301194
- Zhu JH, He F, Liu T, Zhang WH, Liu Q, Yang C (2018) Structural topology optimization under harmonic base acceleration excitations. Struct Multidiscip Optim 57(3):1061–1078. https://doi.org/10.1007/s00158-017-1795-0